Author’s response to reviews

Title: Effectiveness of Music Therapy for Alleviating Pain During Haemodialysis Access Cannulation for Patients Undergoing Haemodialysis: A Multi-facility, Single-blind, Randomised Controlled Trial

Authors:
Masatsugu Kishida (mkishida@ncvc.go.jp)
Yosuke Yamada (yamada19860603@yahoo.co.jp)
Emi Inayama (169676@gmail.com)
Mineaki Kitamura (minekitamura@nagasaki-u.ac.jp)
Tomoya Nishino (tnishino@nagasaki-u.ac.jp)
Keiko Ota (ota.keiko@med.osaka-cu.ac.jp)
Ayumi Shintani (ayumi.shintani@gmail.com)
Tatsuyoshi Ikenoue (tatuyoshi13ikenoue@gmail.com)

Version: 2 Date: 01 Oct 2019

Author’s response to reviews:

As suggested by reviewer #1, please change the word "excruciating" (p6, line 72) to "severe" pain.

<reply>
Thank you for your suggestion.
We have revised the manuscript following the editor's and the reviewer's suggestion (p6,line 72)
The sentence is highlighted in light blue.

In addition, we have removed the Ethical Approval Document and the Funding Documentations in accordance with the Submission Guidelines of Trials.